Author’s response to reviews

Title: The Nicaraguan Pediatric Influenza Cohort Study: Design, Methods, Use of Technology, and Compliance

Authors:

Aubree Gordon (gordonal@umich.edu)
Guillermina Kuan (gkmontes@icsnicaragua.org)
William Aviles (waviles@icsnicaragua.org)
Nery Sanchez (nsanchez@icsnicaragua.org)
Sergio Ojeda (sojeda@icsnicaragua.org)
Brenda Lopez (blopez@icsnicaragua.org)
Lionel Gresh (lionelgresh@gmail.com)
Angel Balmaseda (abalmaseda@minsa.gob.ni)
Eva Harris (eharris@berkeley.edu)

Version: 1 Date: 06 Oct 2015

Author’s response to reviews:

Response to Editorial revisions: 1) Please can you remove the cover letter before resubmitting your manuscript. Thank you, we have removed the cover letter. 2) Please add the title "Background" to the section starting on page 3. We have added the title Background. 3) Please place the Figures and Tables after the References section- ensure that the Figures are uploaded separately and as the images themselves (place the Figure legends after the References) We have moved the Figures and Tables to after the references. 4) Does the informatics system Dashboard have a web address - if so please provide this at first mention. The Dashboard is web-based, but does not have a public access web address.